Author’s response to reviews

Title: TRM2A expression is prognostic in her2+ breast cancer

Authors:

David G Hicks (David_Hicks@urmc.rochester.edu)
Bagi R Janarthanan (Bagi.Janarthanan@roswellpark.org)
Ramya Vardarajan (Ramya.Vardarajan@RoswellPark.org)
Swati A Kulkarni (swati.kulkarni@roswellpark.org)
Thaer Khoury (Thaer.Khoury@roswellpark.org)
Daniel Dim (Daniel.Dim@roswellpark.org)
G. Thomas Budd (BUDDG@ccf.org)
Brian J Yoder (yoderbj@gmail.com)
Raymond R Tubbs (tubbsr@ccf.org)
Marshall T Schreeder (mts@ccihsv.com)
Noel C Estopinal (NRMCT@aol.com)
Rodney A Beck (rbeck@applied-genomics.com)
Yanling Wang (ywang@applied-genomics.com)
Brian Z Ring (bzring@applied-genomics.com)
Robert S Seitz (rseitz@applied-genomics.com)
Douglas T Ross (dross@applied-genomics.com)

Version: 5 Date: 1 February 2010

Author’s response to reviews:

Dear Sir/Madam,

I apologize for the error in submission of a version without conflict of interest and author contribution statements. I have re-submitted the correct file.

Again, my apologies.

Doug Ross